Author’s response to reviews

Title: Evaluation of the impact of a brief educational message on clinicians’ awareness of risks of ionising-radiation exposure in imaging investigations: a pilot pre-post intervention study

Authors:

Ben Young (ben.young@nottingham.ac.uk)
Jo Cranwell (j.cranwell@bath.ac.uk)
Andrew Fogarty (andrew.fogarty@nottingham.ac.uk)
Rob Skelly (rob.skelly@nhs.net)
Nigel Sturrock (nigel.sturrock1@nhs.net)
Mark Norwood (mark.norwood@nhs.net)
Dominick Shaw (dominic.shaw@nottingham.ac.uk)
Sarah Lewis (sarah.lewis@nottingham.ac.uk)
Tessa Langley (tessa.langley@nottingham.ac.uk)
Peter Thurley (peter.thurley2@nhs.net)

Version: 1 Date: 24 Oct 2019

Author’s response to reviews:

Dear Dr Hider,

Many thanks for your assessment of our manuscript and the comments by the reviewers. We are pleased to submit a revised manuscript with changes tracked. For ease of review we have uploaded a separate document that provides a response to each point raised, describes what amendments have been made and where in the revised manuscript these can be found.

Best wishes,

Ben Young